Author's response to reviews

Title: A European perspective of Salmonella enterica subps. arizonae: a case report.

Authors:

Stefano Di Bella (stefano932@gmail.com)
Alessandro Capone (alessandro.capone@inmi.it)
Eugenio Bordi (eugenio.bordi@inmi.it)
Emma Johnson (emmas@hotmail.com)
Maria Musso (maria_musso@hotmail.it)
Simone Topino (simone.topino@inmi.it)
Pasquale Noto (pasquale.noto@inmi.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)

Version: 4 Date: 6 April 2011

Author's response to reviews: see over
Dear Editor:

Attached please find the text of our manuscript, revised according to the Journal style.

Do not hesitate to contact me for further details

Kind Regards

Stefano Di Bella

Stefano Di Bella, M.D.
National Institute for Infectious Diseases
“Lazzaro Spallanzani” – Via Portuense, 292-Rome, Italy
Tel +39 0655170432 fax +39 0655170486
e-mail stefano.dibella@inmi.it
Reply to Editorial Support Assistant:

- Remove the date 'August, 2010' from the case presentation, as it is not relevant to the case report.

  Reply: we removed it.

- Please include a competing interests section at the end of the manuscript, before the reference list. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

  Reply: we included it in the text.

- Please restructure the Abstract into the following three sections: Introduction, Case presentation, and Conclusion. The abstract should be no longer than 350 words.

  Reply: the abstract was restructured as suggested.

- Please clarify whether the tables contained within the two additional files are original to their manuscript. If not, please clarify as to whether they have permission to use this information and kindly clarify the sources of this information.

  Reply: we confirm that the two additional files are original to the manuscript.